Author's response to reviews

Title: Prognostic survival model for people diagnosed with invasive cutaneous melanoma

Authors:

Peter D Baade (peterbaade@cancerqld.org.au)
Patrick Royston (j.royston@ucl.ac.uk)
Phillipa H Youl (PipYoul@cancerqld.org.au)
Martin A Weinstock (martin_weinstock_md@brown.edu)
Alan Geller (AGELLER@hsph.harvard.edu)
Joanne F Aitken (JoanneAitken@cancerqld.org.au)

Version: 4
Date: 13 January 2015

Author's response to reviews:

Thank you for your email of the 9th January, 2015.

1. Please include a competing interest section in your manuscript.

I have moved the competing interest section to the end of the manuscript as per the author's guidelines.

2. Please include a titled conclusion section, as per our journal

I have included a titled conclusion section, as requested.